Dear Sirs,

We hereby submit the revised version of our work titled "Do not forget Diet and Exercise during Ramadan" [Manuscript NO: 88175], following the Reviewer's suggestions and the comments from the Company’s Editor-in-Chief.

Sincerely yours,

Dr Ioannis Ilias
Responses to the Reviewer’s comments and to Editorial comments for MS No: 88175

We thank the Reviewer and the Editors for their comments/suggestions.

Reviewer #1:

Language Quality: Grade B (Minor language polishing)

In the Revised version of the manuscript we made minor language corrections as suggested by the Reviewer.

Editorial Comments

In the Revised version of the manuscript we made corrections to the Table, as suggested by the Company Editor-in-Chief, per the Journal's instructions.